Author's response to reviews

Title: Effectiveness of intensive group and individual interventions against smoking in primary health care settings

Authors:

Maria Ramos (mramos@dgsanita.caib.es)  
Joana Ripoll (jripoll@ibsalut.caib.es)  
Teresa Estrades (testrades@ibsalut.caib.es)  
Isabel Socías (imsocias@ibsalut.caib.es)  
Antonia Fe (afe@ibsalut.caib.es)  
Rosa Duro (reduro@ibsalut.caib.es)  
Maria José González (mjgonzalez@ibsalut.caib.es)  
Margarita Servera (mserveragornals@ibsalut.caib.es)

Version: 3 Date: 17 June 2009

Author's response to reviews: see over
Srs. Editors BMC Public Health:

We have completed the registration process in Current Controlled Trials Ltd of the study “Effectiveness of intensive group and individual interventions against smoking in primary health care settings” (MS: 8503381042784032), so we have attached the ISRCTN number in the application form and in the abstract.

Yours sincerely,

Maria Ramos